SAVINGS IN TRANSNATIONAL HOUSEHOLDS: A FIELD EXPERIMENT AMONG MIGRANTS FROM EL SALVADOR

Nava Ashraf, Diego Aycinena, Claudia Martínez A., and Dean Yang*

Abstract—We implemented a randomized field experiment that tested ways to stimulate migrants' savings in their origin country. We find that migrants value opportunities to exert greater control over financial activities in their home countries. We offered U.S.-based migrants bank accounts in El Salvador, randomly varying migrant control over El Salvador–based savings by offering different accounts across treatments. Migrants offered the greatest degree of control accumulated the most savings. Impacts likely represent increases in total savings; there is no evidence that savings increases were simply reallocated from other savings mechanisms. Enhanced control over home country savings does not affect remittances sent home.

I. Introduction

A TTEMPTS to understand the extent and nature of conflict among household members are central to research on the economics of the family. Many empirical studies have cast serious doubt on the unitary model of the household—the proposition that the joint actions of a household comprising separate optimizing individuals can be represented as the actions of a single utility-maximizing agent.¹ Models that take explicit account of potential preference differences among household members include Manser and Brown (1980), McElroy and Horney (1981), and Lundberg and Pollak (1993). Browning and Chiappori (1998) provide empirical evidence rejecting the unitary model but in favor

* Ashraf: Harvard Business School; Aycinena: Francisco Marroquin University; Martínez A.: Pontificia Universidad Catolica de Chile; Yang: University of Michigan.

This paper was previously titled "Remittances and the Problem of Control." We thank the core members of the project team at ESSMF (Angela Gonzalez, Michelle Guevara, Ronald Luna, Amaris Rodriguez, and Eric Rubin), at FUSADES (Margarita Sanfeliu and Mauricio Shi), and at Banco Agricola (Hector Aguiar, Gustavo Denys, Carla de Espinoza, Mauricio Gallardo, Oscar Hernandez, Sabina Lopez, Ernesto Magana, Katya O'Byrne, and Paul Ponce). We greatly appreciate the collaboration of Enilson Solano and the El Salvador embassy in Washington, D.C. We received valuable feedback and suggestions from Manuel Agosin, Natasha Bajuk, Catia Batista, Charlie Brown, Michael Clemens, Angus Deaton, Esther Duflo, Suzanne Duryea, Jon Guryan, Ricardo Hausmann, Gabriela Inchauste, Takatoshi Kamezawa, Michael Kremer, Steve Levitt, John List, Adriana Lleras-Muney, Ernesto Lopez-Cordova, Osmel Manzano, Doug Massey, Margarita Mooney, Hugo Ñopo, Chris Paxson, Alejandro Portes, Jesse Rothstein, Jesse Shapiro, Ernesto Stein, Bryce Millett Steinberg, Mel Stephens, Don Terry, Steve Wilson, Viviana Zelizer, and participants in several seminars. Alejandra Aponte, Fernando Balzaretti, Sebastian Calonico, Carly Farver, Andres Shahidinejad, and Cristian Sanchez provided excellent research assistance. Martínez A. thanks University of Chile, where this research was partly undertaken. This research was made possible by financial support from the John D. and Catherine T. MacArthur Foundation, the Inter-American Development Bank, the National Science Foundation, the Multilateral Investment Fund, the Empowerment Lab at Harvard University's Center for International Development, and the University of Michigan's International Policy Center. D.Y. acknowledges research support from National Science Foundation award SES-0851570.

A supplemental appendix is available online at http://www.mitpress journals.org/doi/suppl/10.1162/REST_a_00462.

¹ See the review in Strauss and Thomas (1995), as well as Duflo (2003), Rangel (2006), and Martinez (2013).

of household efficiency in resource allocation. Evidence of productive inefficiencies in intrahousehold allocation, however, has been found in a variety of contexts.²

A leading candidate explanation for observed inefficiencies is asymmetry of information in the household, which reduces the ability of household members to enforce mutually beneficial cooperative agreements among themselves.³ This idea has motivated new research that focuses on transnational households, or households with members who have migrated to other countries. Due to the absence of the migrant member, these are households where information asymmetries should be particularly pronounced. If migrants do not share the same financial objectives as family members remaining back home, information asymmetries may prevent migrants from achieving objectives that require the assistance or participation of relatives remaining in the home country.⁴

An improved understanding of financial decision making within transnational households is important because flows of resources within such households are large in magnitude and therefore may have important aggregate impacts. In 2011, migrant remittances sent to developing countries amounted to US\$353 billion. In comparison, in that year, developing country receipts of foreign direct investment (FDI, the largest type of international financial flow going to the developing world) were not quite double that figure (\$646 billion), while receipts of official development assistance (foreign aid) came in a poor third to remittances and FDI, amounting to just \$141 billion.⁵ International financial institutions and developing country governments are keenly interested in identifying policies that can enhance the development impacts of international migration and the accompanying resource flows.⁶

The substantial policy interest in remittances stands in stark contrast to the limited empirical evidence that can help guide policy.⁷ A number of questions related to deci-

² See Udry (1996), Dercon and Krishna (2000), Goldstein, de Janvry, and Sadoulet (2005), and Dubois and Ligon (2005), among others.

⁴ In analyses of observational data, Chen (2006) and De Laat (2008) find empirical patterns consistent with asymmetric information in migrant households, as evidenced by migrant monitoring of spouses in home areas (among domestic migrants in Kenya and China, respectively).

⁵ Data on remittances, FDI, and ODA are from World Development Indicators 2013.

⁶ Recent reports on remittances funded by the Inter-American Development Bank include Pew Hispanic Center (2002) and Terry and Wilson (2005). World Bank publications include World Bank (2006, 2007).

⁷ See Yang (2011) for a review of the state of research on the economics of migrant remittances.

Received for publication December 22, 2011. Revision accepted for publication March 14, 2014. Editor: Gordon Hansen.

³ Ashraf (2009) shows that individual saving decisions change when observed by one's spouse. Recent work on the savings and risk-sharing consequences of intrahousehold preference differences and asymmetric information includes Schaner (2011), Kinnan (2011), and Hertzberg (2011).

sion making in transnational households are of general economic interest and policy relevant. To what extent do migrants seek to monitor and control financial decision making by household members remaining in the home country? What kinds of financial product innovations might enhance migrant ability to exert such monitoring and control? If given the opportunity to do so, would migrants seek to exert greater control over such decisions, and what would be the resulting impacts on financial decision making in the transnational household?

To shed light on these questions, we conducted a randomized controlled trial among U.S.-based migrants from El Salvador. We randomized offers to migrants of financial products that varied the degree to which they could monitor and control savings in bank accounts in their home country.⁸ In survey data we collected, Salvadoran migrants report that they would like recipient households to save 21.2 percent of remittance receipts, while recipients prefer to save only 2.6 percent of receipts. Migrants often intend savings to be for future use by the recipient household, but such savings can also be intended for the migrants themselves. In the latter case, migrants may send their own funds to be saved in El Salvador because they perceive savings held in the United States to be relatively insecure (particularly for undocumented migrants who fear deportation and loss of their assets).

Migrants in the study were randomly assigned to a control group or to one of three treatment conditions that offered financial products with varying levels of monitoring and control over savings in El Salvador. We examine the effect on our outcomes of interest: take-up and balances in savings accounts of various types. Our comparison group, referred to as treatment 0, received no offer of any new financial products. In treatment 1, migrants were offered the opportunity to open a new account in the name of someone in El Salvador, into which the migrant could remit funds. This account allows the migrant to deposit but not to withdraw or to observe withdrawals. Treatment 2 offered the migrant the opportunity to open an account to be held jointly by the migrant and someone in El Salvador. This new joint account allows joint observability of account balances as well as joint withdrawals (both the migrant and the El Salvador person are given an ATM card for the account). Finally, in treatment 3, migrants were offered, in addition to the accounts offered in treatments 1 and 2, the option to open an account in the migrant's name only.⁹ Thus, each treatment nests the one prior to it so that the effect of offering additional products can be understood. Project staff delivered a marketing pitch for each product according to its features.¹⁰ Data on financial transactions at our partner bank come from the bank's administrative records. Baseline and follow-up surveys administered to migrants in the United States provide data on other outcomes.

Our results provide evidence that migrants do value and take advantage of opportunities to exert control over savings in their home country. Migrants were much more likely to open savings accounts at the partner bank in El Salvador and accumulated more savings at the partner bank if they were assigned to the treatment condition offering the greatest degree of monitoring and control (treatment 3). Migrants desire savings that are jointly held with family members, as well as savings only for themselves. We observe substantial increases in savings in both the joint accounts shared between migrants and someone in El Salvador (offered in treatments 2 and 3) and in the accounts for migrants alone (offered only in treatment 3).¹¹ This increase in savings in the new accounts we offered is likely to be a true increase in savings; we find no evidence that these funds were simply shifted over from other types of savings (either from other accounts at the partner bank or from other types of savings reported in the follow-up survey).

Strikingly, the impact of treatment 1 (where we offered accounts in the name only of someone in El Salvador) on savings was much smaller in magnitude and not statistically significantly different from 0. This result is also important, as it reveals that the frequently made policy recommendation to foster savings in migrants' home countries by encouraging migrants to remit directly into savings accounts of remittance recipients would be much less effective compared to interventions that also improved and encouraged migrant monitoring and control over home country savings.

We also provide additional evidence suggestive that the increases in savings due to treatment 3 are due to improvements in migrant ability to control recipient savings in El Salvador. We show that savings increases in joint accounts at the partner bank (shared by migrants and someone in El Salvador) are concentrated among migrants who revealed a demand for control over remittance uses in the baseline (pretreatment) survey (e.g., among migrants who had previously sent funds to El Salvador for others to administer or who were aware of disagreements between migrants and recipients over the use of remittances).

In addition, although both treatments 2 and 3 offered joint accounts shared by migrants and someone in El Salva-

⁸ Chin, Karkoviata, and Wilcox (2010) conduct an experiment on savings among Mexican immigrants in Texas, finding that immigrants are more likely to open U.S. savings accounts, accumulate more savings in the United States, and remit less to Mexico when they are helped to obtain an identification that facilitates opening U.S. bank accounts.

⁹ In treatments 2 and 3, upon request migrants would also have been allowed to open an account for someone in El Salvador only (the account offered in treatment 1). No migrants made such a request.

¹⁰ Moving from treatments 0 to 3, marketing pitch content was only added (never subtracted), so the marketing pitches were nested in the same way that the product offers were.

¹¹ These impacts are large in economic magnitude. For example, treatment 3 leads to an increase of \$285 in average account balances (across all accounts at the partner bank) in the twelve months posttreatment. By comparison, average balances at the partner bank were just \$183 in the comparison group (treatment 0) over the same period.

dor, take-up of these accounts was statistically significantly higher in treatment 3 when migrant-only accounts were also offered. This pattern is suggestive of decoy effects (Laran, Dalton, & Andrade, 2011; Chatterjee & Rose, 2012). Offering migrant-only accounts as part of the menu of products may have drawn attention to the control features of accounts offered. The joint account, while not allowing the same degree of full control as the migrant-only account, provided greater control than most accounts that migrants remit into: it provided the migrant the opportunity to check balances and an ATM card with which to withdraw from the account. Offering the migrant-only account alongside the joint account in treatment 3 may have encouraged migrants to pay more attention to the control features of the joint account.

We also provide evidence suggesting that our treatment effects do not derive from the marketing pitches alone. Joint account savings at other banks (aside from our partner bank) are not affected by the treatments. We interpret this as evidence that our offer of accounts at our partner bank was necessary to produce the effects on savings we observe.

The remainder of this paper is organized as follows. Section II provides details on the study design. Section III describes the characteristics of the sample. Section IV presents the main empirical results. Section V provides discussion and additional analyses. Section VI concludes.

II. Study Design

A. Sampling Protocol and Baseline Survey

Study participants are immigrants in the greater Washington, D.C., area. To be eligible for inclusion in the sample, individuals had to have been born in El Salvador, entered the United States for the first time within the last fifteen years, and sent a remittance to someone in El Salvador within the last twelve months.

To recruit migrants, we stationed our survey team at the two Salvadoran consulates in the Washington, D.C., area (in Washington, D.C., and in Woodbridge, Virginia). The main services study participants sought at the consulate were passport renewals, civil registration (of births, deaths, and marriages), and assistance with processing of temporary protected status (a special provision allowing temporary legal work for Salvadorans and those other nationalities who entered the United States after natural disasters or civil strife in the home country). The consulate of El Salvador serves Salvadorans regardless of their legal status, and so the sample likely includes both documented and undocumented migrants (we intentionally did not ask any questions related to immigration status.) The El Salvador consulate endorsed our study; intermittently, a consular staff member would make an announcement in the waiting area endorsing participation in the study.

Survey team members were mostly of Salvadoran origin and mostly women. Survey team members approached individuals in the waiting area of the consulate, inviting them to participate in the study. The D.C. baseline survey fieldwork ran from June 2007 to January 2008. Individuals were told the name of the study, the academic institutions involved, and that the survey was about "Salvadorans who send remittances and their family who receive them in El Salvador." Individuals were asked the three screening questions (above), and those who were eligible were invited to participate. Those who agreed and signed consent forms were administered the baseline survey at the consulate. Of 5,288 people approached at the consulate, 3,611 passed the screening questions, and 1,986 agreed to participate and completed the baseline survey.

After completion of a migrant baseline survey in the D.C. area, a separate survey team attempted to survey the individual in El Salvador whom the migrant identified as his or her primary remittance recipient (PRR). The survey team successfully completed 1,338 El Salvador household surveys between November 2007 and June 2008. After attempting a survey of an El Salvador household (whether successfully completed or not), a project staff member (a "marketer") in D.C. then attempted to schedule (by phone) and carry out (in person) a marketing visit with the corresponding migrant, at which the treatments were administered. Many migrants were reluctant to make time for these visits, and we were unable to recontact some respondents due to invalid or changed phone numbers. Marketers made appointments for 1,054 marketing visits. Due to no-shows at the scheduled visits, our final sample size is slightly lower.

Marketers carried out marketing visits at locations chosen by migrants. Random assignment to treatment 0, 1, 2, or 3 occurred only after a marketing visit with the migrant had been scheduled but before the actual visit was carried out. Study participants did not learn their treatment assignment prior to the actual face-to-face marketing visit. Visits took from one to two hours. Marketers were paid a flat fee for each completed visit that was the same for all treatment conditions (to avoid any differential incentive to complete visits of different types). The marketing visits were carried out between December 2007 and July 2008. The four treatments are described below, and details of marketing scripts are in online appendix A.

Our sample for analysis in this paper is 898 migrants with whom a face-to-face marketing visit was successfully carried out (along with their associated primary-remittancerecipient household in El Salvador). While attrition from the baseline survey to completed marketing visits is certainly detrimental to our sample size, it should not affect the internal validity of the study, because at no stage prior to the in-person marketing visit did study participants know their treatment assignment.

The resulting migrant sample comprises a reasonable cross-section of Salvadoran migrants in the Washington,

D.C., area. Online appendix table 1 presents means of several key baseline variables for observations in our baseline data (column 1), in comparison with corresponding means for Salvadoran-born and Hispanic individuals in the U.S. Census 2000 in the Washington, D.C., metropolitan area, separately for men and women. While differences are not dramatic, there are some key differences between our sample and U.S. Census Salvadorans in the D.C. metropolitan area. Focus for the moment on the comparison with column 2, for all Salvadoran-born individuals regardless of U.S. citizenship. Our sample is more male, at 71 percent versus 57 percent. Our sample has also arrived somewhat more recently in the United States, with 49 percent and 51 percent of men and women, respectively, having been in the United States for five years or less at the time of survey, compared to corresponding figures of 33 percent and 29 percent for U.S. Census Salvadorans. Our sample is slightly more educated: 30 percent and 36 percent of our sample men and women, respectively, have a high school diploma or more, compared to 27 percent and 30 percent of U.S. Census Salvadoran men and women, respectively. Our sample is less likely to have U.S. citizenship, at 0 to 1 percent, compared to 10 to 12 percent for U.S. Census Salvadorans. Finally, our sample is more likely to be married or partnered, at 53 percent and 73 percent of men and women, respectively, compared to 45 percent and 57 percent for men and women U.S. Census Salvadorans, respectively. These differences in relation to our baseline sample are quite similar when restricting the sample of U.S. Census Salvadorans to those without U.S. citizenship (column 3) and when examining Hispanics without U.S. citizenship (column 4).

B. Experimental Design

In conjunction with our partner bank in El Salvador, Banco Agricola, we designed the savings facilities offered in this project "Cuenta Unidos" and "Ahorro Directo" (described further below). Neither of these savings products existed previously. Our study offered the new products only to study participants randomized into certain treatment conditions. That said, anyone asking for these new products (say, if they heard about them from study participants) were allowed to open them by partner bank staff. To reduce contamination of our treatment effect estimates from spillovers to the comparison group (treatment 0), our partner bank agreed not to market or advertise the new products designed for this study in any fashion until the follow-up survey was implemented (roughly a year after treatment).

Migrants were randomly assigned to one of three treatment groups or a comparison group, each with equal (25 percentage) probability. We randomized after stratifying migrants into 48 cells representing unique combinations of four baseline categorical variables: gender (male, female), U.S. bank account ownership (yes, no), primary remittance recipient's relationship to migrant (parent, spouse, child, other), and years in United States (0–5, 6–10, 11–15). Stratified randomization was carried out between the completion of the baseline survey and the marketing visit attempt; because not all marketing visits were successful, it is not guaranteed that treatment conditions are precisely balanced on the stratification variables.

Treatments were administered in the marketing visits. Migrants in the comparison group (treatment 0) were not offered any new products. (Because this study investigates control over savings, to avoid confusion we refer to treatment 0 as the "comparison group," not the "control group.") The three treatment groups were labeled 1, 2, and 3. The presence of the comparison group allows us to observe outcomes for a comparable sample where none of the products were offered.

To help track migrants' remittance behavior after the visit, all visited migrants were given a special card (called a "VIP card") that provided a discount for sending remittances via Banagricola remittance locations in the D.C. area. Banco Agricola's normal remittance charge is \$10 for a remittance up to \$1,500, and the VIP card allowed the migrant to send a remittance for a randomly determined price of either \$4, \$5, \$6, \$7, \$8, or \$9 (once randomly assigned at the outset, the price was fixed for the validity period of the card).¹² Eligibility for the card was conditional on the migrant's presenting an identification document of some sort (usually a Salvadoran passport). Migrants were told to bring an identification document in the initial phone call making the appointment for the marketing visit.

C. Treatment Groups

Treatment 0 (comparison group): Encouragement to remit into bank account of someone in El Salvador. Migrants assigned to treatment 0 were visited by a marketer who encouraged them to remit into El Salvador bank accounts. Marketers emphasized the benefits of remitting funds directly into accounts and of remittance-recipient access to funds via ATM or debit cards (rather than having to wait in a teller line to receive a remittance). Migrants were offered the VIP card but were not offered any new savings facilities. This generic pitch to remit into bank accounts was included in the control condition to ensure that any increases in savings seen in treatments 1, 2, or 3 (versus corresponding changes in treatment 0) were not due simply to the encouragement provided by the marketers to remit into bank accounts in El Salvador.

¹² This remittance price randomization was independent of the randomization into treatments 0, 1, 2, or 3 and so does not confound interpretation of any differences across treatments. In addition, migrants did not learn the actual discounted VIP price until after the marketing visit had concluded. The remittance price randomization was implemented for a separate study within the same study population on the impact of remittance prices on the frequency and amount of remittances (Aycinena, Martinez, & Yang, 2010).

Treatment 1: Offer of account for someone in El Salvador. In treatment 1, marketers also emphasized the same benefits of remitting into bank accounts (as in treatment 0) and provided the VIP card. But unlike in treatment 0, in treatment 1, this was combined with an offer of assistance in setting up an account in the name of someone in El Salvador, into which the migrant could remit. While the migrant would be able to make deposits into this account, he or she would not be able to observe the balance of and withdrawals from this account. Relative to treatment 0, the treatment 1 marketing pitch also added a brief comment that "savings for your remittance recipient in El Salvador" was a benefit of the treatment 1 offer (but with no other elaboration on the general benefits of bank accounts). To equalize account opening costs across treatments 1 and 2, this remittance-recipient-only account offered in treatment 1 was exactly the same product (Cuenta Unidos) offered in treatment 2. The difference was that in treatment 1, we did not facilitate making the migrant a joint account holder on the Cuenta Unidos account. Migrants could identify anyone in El Salvador as the account holder (not just the PRR to whom the baseline survey was administered.) If migrants were interested, they filled out forms to provide the name, address, and phone number of the individual in El Salvador for whom the account was intended. The marketer offered to let the migrant use a project cell phone to call the person in El Salvador during the visit to inform him or her of the new account.¹³ Within the next few days, project staff arranged by phone for the individual in El Salvador to meet with the branch manager of the nearest Banco Agricola branch in El Salvador to complete the final account opening procedures in person.

Effects of treatment 1 on take-up and savings accumulation (in relation to treatment 0) would reflect the impact of offering assistance with account opening procedures. In addition, relative to treatment 0, treatment 1 potentially improves what one might call the identity precision of remittances and savings: the migrant's ability to channel remittances toward a particular person's savings account. Because the account offered in treatment 1 is in the name of someone in El Salvador, any impacts found could not be due to changes in the migrant's ability to monitor or control savings balances. Even if it failed to offer migrants greater monitoring or control, migrants might have found the account offered in treatment 1 attractive if they thought that a savings account would be beneficial for the recipient or if they wanted to use a recipient's savings account as a safe and convenient destination for remittances to that recipient.

Treatment 2: Offer of joint account for migrant and someone in El Salvador. In treatment 2, we offered migrants the Cuenta Unidos account, which was newly designed for this project. This savings facility allows joint ownership by both an individual in El Salvador and the migrant in the United States. Joint account owners in both the United States and El Salvador had ATM cards and full access to account information. Migrants could deposit funds into the account via remittances, withdraw with their ATM card at U.S. ATMs, and check the balance on the account by calling a toll-free U.S. telephone number. Joint account owners in El Salvador could deposit and withdraw using their ATM cards or going to a bank teller.

The substantive content conveyed by the marketing pitch in treatment 1 was also conveyed in treatment 2; in addition, the treatment 2 marketing pitch also noted that both the migrant and the El Salvador account holder could verify the balance on the Cuenta Unidos account and that the migrant could withdraw funds from the account from the United States.

If migrants were interested in Cuenta Unidos, they filled out account opening forms. As in treatment 1, migrants provided contact information for the joint account holder in El Salvador, and marketers and other project staff facilitated the account opening process on the El Salvador side (by offering the migrant a free call on their project cell phone and arranging the account opening appointment in El Salvador). Migrants could identify anyone in El Salvador as the joint account holder. Migrants had the option to not have joint ownership of the new account (in other words, they could replicate the account offered in treatment 1).¹⁴ Compared to treatment 1, treatment 2 offered migrants the ability to monitor the savings of family members but did not provide full control over the funds. The joint account holder in El Salvador had complete freedom to withdraw the entire savings balance from the account.

Treatment 3: Offer of joint account for migrant and someone in El Salvador, plus migrant-only account. Treatment 3 nests treatment 2 while adding an additional savings facility: an account exclusively in the migrant's name, known as Ahorro Directo (also newly designed for the project) This is an account only in the name of the migrant. The migrant could deposit into the account by remitting into it and received an ATM card for withdrawals at U.S. ATMs.¹⁵

In this marketing visit, Cuenta Unidos and Ahorro Directo were offered to the migrant in sequence. Cuenta Unidos was offered first, using a marketing script identical to the one used for treatment 2. The marketing script for

¹³ To mitigate any possibility that talking to the primary recipient might have an effect on outcomes of interest, migrants assigned to treatment 0 were also offered a complimentary phone call to the primary recipient from the project cell phone.

¹⁴ However, all accounts we assisted in opening in treatment 2 were joint accounts. Not once did a migrant request to forgo joint ownership of Cuenta Unidos in treatment 2.

¹⁵ A question that arises is why migrants would not have opened their own bank accounts in El Salvador prior to being offered them by our project. Our results will indeed show that the marketing and account opening assistance offered by our project led to opening of these accounts. While our study is not designed to shed light on these barriers directly, it is likely that the reduction in transaction costs of account opening due to our account opening assistance was nontrivial.

Ahorro Directo, which followed, emphasized its usefulness for exclusive control over funds, since the account would not be shared with anyone else. The script noted that no one other than the client would be able to check account balances, have access to the account, or even know of the existence of the account. The script also noted that no intermediaries (e.g., family members) would be needed for the client to save in El Salvador. In addition, the script noted the benefit of improved security if visiting El Salvador by reducing the need to carry large amounts of cash.¹⁶

For the purpose of the study, it is important to be able to rule out that any differences across treatments 2 and 3 are due to differences in transaction costs. Therefore, in treatment 3, if migrants wanted to open an Ahorro Directo account, we required them to also open a Cuenta Unidos account, ensuring that account opening transaction costs were identical across treatments 2 and 3.¹⁷ In addition, migrants were allowed to open an account only in the name of a beneficiary in El Salvador (as in treatment 1) if they requested it.¹⁸

In sum, treatment 3 offered the migrant the greatest ability to control funds in savings accounts in El Salvador, unlike treatment 2 where ownership had to be joint with someone else. The difference in takeup and savings between treatments 2 and 3 therefore reveals the incremental impact of offering migrants the ability to exclusively control their El Salvador savings balances.

Other important notes on the treatment conditions. It is important to be clear that the pitch for each product did not vary across treatment conditions. However, because they differed in the products offered, treatments did involve different pitches that were delivered alongside their associated products. In particular, the joint account, Cuenta Unidos, was offered in both treatments 2 and 3, but the pitch that

¹⁷ By requiring that migrants wanting an Ahorro Directo also open a Cuenta Unidos, the migrant had to get an individual in El Salvador to physically visit a Banco Agricola branch there to fill out account opening documents. If we had not instituted this requirement, the transaction cost for opening an Ahorro Directo would have been much lower than for opening a Cuenta Unidos, because the former would not have required a trip by someone in El Salvador to a Banco Agricola branch. The upshot of this design is that take-up of Ahorro Directo in treatment 3 will be a lower bound of what take-up would have been had we not instituted this requirement. We judged that improving clarity of interpretation was worth the sacrifice of potentially lower take-up in treatment 3. Note that in treatment 1, the individual in whose name the account was opened also had to go to a branch in El Salvador, so transaction costs are also equalized with treatment 1.

¹⁸ Again, as in treatment 2, no migrant assigned to treatment 3 who chose to open a Cuenta Unidos account opted to forgo joint ownership.

accompanied that product offer was identical across those two treatments. In treatment 3, Ahorro Directo was also offered, with its own additional pitch. Online appendix A provides details on these product-specific pitches.

D. Outcome Variables

The primary outcome variables we examine are savings balances of various types at the partner bank, which we obtained from the partner bank's administrative databases. We focus on savings balances over the 12 post treatment months (after the study participant's marketing visit), but we also provide estimates of treatment effects up to 48 months post treatment. To obtain these data, both the migrant and his or her PRR were located in the partner bank's administrative data by a search on the basis of matching variables (given name, surname, address, phone number, and age) that were obtained from study participants before treatment assignment. The search was performed by bank staff on the basis of our instructions. We provided bank staff with the matching variables to ensure that no additional identifying data (such as improved addresses and phone numbers) that might have been obtained from individuals taking up products in the treatment groups would be used in the matching process. Had we not done so, our treatment effects on bank balances would be biased upward because more individuals might have been successfully found in the treatment groups that involved take-up of bank products.¹⁹

It should be noted that this matching procedure locates only accounts owned by either the migrant or PRR (or both). In other words, it locates joint accounts shared by migrants and PRRs, joint accounts shared by migrants and individuals other than the PRR, accounts held only by the PRR (without migrant co-ownership), and accounts held only by the migrant (without PRR co-ownership). However, the procedure fails to locate accounts that migrants might have opened in the name of non-PRRs alone (without migrant co-ownership) because we do not have a pretreatment list of potential non-PRR account holders to search for in the database. To the extent that migrants did open accounts in the name of non-PRRs alone, our results here will understate the impacts of the interventions on account opening and savings.

In addition, we fielded migrant follow-up surveys roughly one year after the initial product offer (from March to June 2009) to measure impacts on savings outside the partner bank. Follow-up surveys of D.C.-based migrants were conducted by phone by a survey team calling from El Salvador.

¹⁶ Notwithstanding the way Ahorro Directo was marketed, one might imagine that migrants could still have used these accounts for joint savings with El Salvador persons, for example, if migrants sent their Ahorro Directo ATM cards to individuals in El Salvador to provide them account access. There is no evidence that this occurred, however. Analysis of withdrawal data from these accounts shows that transactions on these accounts (both deposits and withdrawals) occurred exclusively on the U.S. side over the period analyzed in this paper. Not a single deposit into or withdrawal from an Ahorro Directo account occurred in El Salvador through the end of 2009.
¹⁷ By requiring that migrants wanting an Ahorro Directo also open a

¹⁹ We have confirmed that the data quality of the matching variables is not differential by treatment status. Further confirmation that the quality of the matching did not vary by treatment status is that prior to treatment, savings balances at the partner bank are balanced across treatment conditions (see appendix table 2).

338

E. Estimation Strategy

Dependent variables of interest in this paper are take-up rates of and balances in savings accounts at the partner bank. Let Y_i be the dependent variable of interest (say, total savings at the partner bank). Let $Z1_i$ be an indicator variable for assignment to treatment 1, $Z2_i$ be an indicator variable for assignment to treatment 2, and $Z3_i$ be an indicator variable for assignment to treatment 3. We estimate treatment effects using the following regression equation:

$$Y_i = \delta + \alpha_1 Z \mathbf{1}_i + \alpha_2 Z \mathbf{2}_i + \alpha_3 Z \mathbf{3}_i + X_i \phi + \mu_i \tag{1}$$

Coefficients α_1 , α_2 , and α_3 are the impact on the dependent variable of treatments 1, 2, and 3, respectively. We focus on intent to treat (ITT) effects and so are estimating the effect of offering (rather than opening) the various accounts.

The difference $(\alpha_3 - \alpha_2)$ represents the difference in the impact of treatment 3 in relation to treatment 2 and the difference $(\alpha_2 - \alpha_1)$ represents the difference in the impact of treatment 2 in relation to treatment 1. X_i is a vector of control variables which include fixed effects (for marketer, stratification cell, and month of marketing visit when treatments were administered) and an indicator variable for the individual expressing demand for control over financial decision making of primary remittance recipients in the baseline survey (described further below). μ_i is a mean-zero error term. For all coefficient estimates, we report robust (Huber/White) standard errors that account for survey design.²⁰

One of our key dependent variables is savings balances, which potentially have large outliers that could have disproportionate influence on the estimates. Therefore, in all results tables for impacts on savings balances, we focus on impacts on the quartic root of savings balances, a specification that reduces the influence of outliers (these will be presented in panel A of the relevant tables). We also report impacts on the dollar amount of savings balances (in panel B) but will consider this specification secondary to the quartic root specification.

III. Sample Characteristics

Our primary sample for analysis, which we use to analyze impacts on savings at the partner bank, consists of the 898 D.C.-area migrants who completed a baseline survey as well as a marketing visit some months later.

A. Characteristics of Migrants and Remittance-Receiving Households

Summary statistics are presented in table 1. Several measures of demand for control are available in the baseline survey administered to migrants. We construct five separate indicator variables equal to 1 (and 0 otherwise) from migrant reports of the following: (a) the migrant had ever paid directly for expenditures of remittance recipients in El Salvador rather than sending cash (7.7 percentage of migrants did so); (b) the migrant had sent funds home for others to administer on his or her behalf (23.7 percentage of migrants did so); (c) the migrant was interested in direct payments to improve control over remittance uses (20.7 percentage of migrants said yes); (d) the migrant knew anyone who had had conflict with recipients over remittance uses (14.6 percentage of migrants said yes); (e) the migrant has had conflict with his or her own remittance recipients over remittance uses (4.9 percentage of migrants said yes). We construct an overall indicator of "demand for control" that takes the value of 1 if the migrant answers affirmatively to any of the five above mentioned indicator variables and 0 otherwise. Fifty-one percent of migrants answered yes to at least one of these questions.

The baseline survey also included three questions assessing financial literacy that have been included in surveys of financial literacy worldwide (Lusardi & Mitchell, 2007).²¹ Sixty-six percent, 64 percent, and 37 percent of migrants responded correctly to the questions on, respectively, compound interest, inflation, and mutual funds. We also asked whether migrants tracked spending and budgeted expenses; 46 percent of migrants reported "always" or "almost always" doing so.

B. Balance and Attrition across Treatment Groups

It is important to check whether randomization across treatments achieved the goal of balance in terms of pretreatment characteristics of study participants. Online appendix table 2 presents the means of a number of baseline variables for each treatment group as reported prior to treatment. The first column of reported *p*-values is for *F*-tests of equality of means across the treatment groups for each variable separately. The other three columns of *p*-values are for *F*-tests of the pairwise equality of means between observations in treatment 0 and, respectively, treatments 1, 2, and 3. The table also examines account ownership and average savings balances of study participants at the partner bank during twelve months prior to the month of the marketing visit (specified as the quartic root and in dollars, in parallel to the regression results to come).

The first nine variables listed in the table are the stratification variables (gender, U.S. bank account, relationship to remittance recipient, and years in United States category).

²⁰ Specifically, regressions are run with Stata's "svy" option, where data are "svyset" to account for survey strata (the stratification cells).

 $^{^{21}}$ The questions are (a) "Suppose that you have \$100 in a savings account with a 2% annual interest rate. If you do not touch the money in this account, how much do you think you will have in five years?" (Options are "less than \$102," "exactly \$102," and "more than \$102"; the correct answer is "more than \$102."), (b) "Imagine that the interest rate in the savings account where you have \$100 is 1%, and that inflation is 2% per year. A year from now, would you be able to buy more, the same, or less than today with the money in the account?" (correct answer is "buy less"). (c) "Do you think that the following statement is true or false? To buy stocks in only one company is more secure than to invest in a mutual fund" (correct answer is "false").

TABLE 1.—SUMMARY STATISTIC

			10th		90th	Number of
	Mean	SD	Percent	Median	Percent	Observations
	intean	55	1 010010	mount	1 0100111	o ober runono
Treatment indicators and stratification variables						
Treatment 0 (no savings facility offered)	0.24	0.43	0	0	1	898
Treatment 1 (remittance recipient account only)	0.23	0.42	0	0	1	898
Treatment 2 (joint account)	0.27	0.45	0	0	1	898
Treatment 3 (joint + migrant account)	0.25	0.43	0	0	1	898
Migrant is female	0.29	0.45	0	0	1	898
Migrant has U.S. bank account	0.63	0.48	0	1	1	898
Recipient is migrant's parent	0.55	0.50	0	1	1	898
Recipient is migrant's spouse	0.11	0.31	0	0	1	898
Recipient is migrant's child	0.04	0.19	0	0	0	898
Recipient is migrant's other relative	0.30	0.46	0	0	1	898
Migrant has been in United States 0–5 years	0.50	0.50	0	0	1	898
Migrant has been in United States 6–10 years	0.40	0.49	0	0	1	898
Migrant has been in United States 11–15 years	0.11	0.31	0	0	1	898
Baseline variables from D.C. migrant survey						
Migrant's years in the United States	5.57	3.60	1	5	11	898
Migrant has El Salvador bank account	0.17	0.38	0	0	1	898
Migrant's annual income (US\$)	30,999	56,292	11,700	24,960	48,822	865
Migrant's household's annual income (US\$)	39,620	87,551	10,530	31,200	65,000	896
Migrant's years of education	8.28	4.33	1	9	12	897
Migrant's age	30.88	7.65	22	30	41	894
Migrant's annual remittances sent (US\$)	4,990	4,124	1,200	3,900	9,600	898
Migrant's total household savings balance (US\$)	2,851	5,111	0	750	8,100	806
Migrant is US citizen	0.007	0.082	0	0	0	894
Migrant household size in United States	4.81	2.15	2	5	8	898
Migrant is married or partnered	0.59	0.49	0	1	1	897
Past experience with direct payments	0.08	0.27	0	0	0	898
Sent funds to El Salvador for others to administer	0.23	0.42	0	0	1	898
Interested in direct payments to increase control	0.21	0.41	0	0	1	898
Aware of disagreements with recipients over remittance uses	0.15	0.35	0	0	1	898
Have had disagreements with recipients over remittance uses	0.05	0.22	0	0	0	898
Demand for control (union of above five indicators)	0.51	0.50	0	1	1	898
Correct answer to compound interest question	0.66	0.47	0	1	1	898
Correct answer to inflation question	0.64	0.48	0	1	1	898
Correct answer to mutual fund question	0.37	0.48	0	0	1	898
Tracks spending and budgets expenses	0.46	0.50	0	0	1	897
Baseline variables from El Salvador household survey						
Recipient's total household savings balance (US\$)	382	1,732	0	0	380	733
Recipient's annual remittances received (US\$)	3,182	2,787	900	2,400	6,000	725
Pretreatment savings at partner bank						
Any savings account in twelve months prior to treatment (indicator)	1.02	2.08	0	0	4	898
Savings balance, average over twelve months prior to treatment-quartic (US\$)	1.02	2.08	0	0	4.45	898
Savings balance, average over 12 months prior to treatment (US\$)	243	1,085	0	0	391	898

Survey data collected from June 2007 to January 2008 among Salvadoran migrants in Washington, D.C., and from November 2007 to June 2008 among households in El Salvador identified as D.C. migrant's primary remittance recipient

The *p*-values on the *F*-test of the joint equality of means across all treatments are all far from conventional significance levels. In only 1 out of 27 pairwise comparisons with the treatment 0 mean is there a statistically significant difference in means (the comparison between treatments 2 and 0 for "recipient is migrant's other relative").²² This one rejection of equality is not worrisome, however, as the regression estimates to come will control for stratification cell fixed effects (estimates will take advantage only of variation in treatment within stratification cell), and all results are robust to inclusion or exclusion of the stratification cell fixed effects.

The remaining variables in the table are other variables for which observations were not stratified prior to treatment assignment. For all these remaining variables, the *p*-values in essentially all cases are also large, and we cannot reject the hypothesis that the means are identical across treatment groups.²³

Follow-up attrition rates across treatments are presented in the bottom row of online appendix table 2.²⁴ The followup survey contains 508 observations with valid migrantreported savings data, for an overall attrition rate of 43.4

²² As mentioned earlier, stratification was carried out prior to the marketing visit, and so failure to complete the marketing visit could have led to imbalance on the stratification variables. In addition, some of the stratification cells had small numbers of migrants. When the number of migrants in a cell was not a multiple of four, it was not possible to assign exactly 25 percentage of migrants within a cell to each treatment.

²³ The three exceptions are the pairwise comparison between treatments 2 and 0 for "migrant's annual remittances sent," "migrant is U.S. citizen," and "migrant is married or partnered," in which cases the means are significantly different at the 10 percent level. This small number of significant differences can be expected to arise by chance even with randomization. ²⁴ Attrition can be due to noncompletion of the follow-up survey, as

well as missing savings data in that survey.

percent. Attrition rates between treatment 1 and treatment 0 are not statistically significantly different from one another (45 percent and 49 percent, respectively). However, observations in treatment 2 have statistically significantly lower attrition rates than treatment 0 (amounting to 10 percentage points lower attrition). In addition, treatment 3 has about 7 percentage points lower attrition than does treatment 0 (and this difference is marginally statistically significant, with a p-value of 0.13).

This pattern of treatment-related attrition raises concerns about selection bias in estimates of treatment effects on outcomes measured in the follow-up survey. In online appendix table 3, we investigate the balance of baseline characteristics across treatment conditions in the follow-up sample (N = 508), analogous to those examined in online appendix table 2. While across most variables there does not seem to be dramatic evidence of differences across treatment conditions in the follow-up sample, there does seem to be a worrying imbalance in pre-treatment savings at the partner bank. An *F*-test rejects equality of the quartic root of savings at the partner bank prior to treatment across all treatment groups at the 10 percent level, and the difference in this variable between treatments 3 and 0 is statistically significant at the 5 percent level.²⁵

Due to these patterns of differential survey attrition and imbalance in the follow-up survey sample, care must be taken in interpretation of any treatment effects estimated using this sample. Our focus, therefore, will be on the outcomes observed in the administrative data from the partner bank, which are not subject to such concerns.²⁶

IV. Impact of Treatments on Savings

In this section we examine the impact of the treatments. We first discuss impacts on account opening and on savings in accounts at the partner bank. We then discuss whether treatment effects are likely to reflect shifting of funds across savings mechanisms, and in that context, we examine treatment effects on savings reported in the follow-up survey.

A. Impact on Account Opening at Partner Bank

We first estimate equation (1) examining the impact of the various treatment conditions on take-up of savings accounts at the partner bank. We regress an indicator for the existence of a certain type of account in the first through twelfth month post treatment on indicators for being assigned to each of treatment conditions 1 through 3.²⁷ We examine three categories of accounts separately, distinguishing between the two types of new accounts designed for this research project ("project accounts") and other accounts at the partner bank:

- Cuenta Unidos accounts. Recall that in treatments 2 and 3, we offered Cuenta Unidos accounts as joint accounts between migrants and someone in El Salvador. In treatment 1, the accounts we offered were also Cuenta Unidos accounts, but in that case we did not offer migrants the opportunity to be joint account holders—the accounts were offered as accounts for individuals in El Salvador alone.²⁸
- 2. *Ahorro Directo accounts*. These accounts were in the name of migrants only.
- 3. *Other accounts* in the name of the migrant or the PRR, excluding those (Cuenta Unidos and Ahorro Directo) offered by our project.²⁹

Coefficient estimates for regression equation (1), where the dependent variable is the existence of a Cuenta Unidos account, are regressions labeled (a) in table 2 (without and with control variables, respectively). The coefficient on the constant term in the first column of regression a indicates that for 5.9 percent of migrant-PRR pairs assigned to the comparison group (treatment 0), a Cuenta Unidos account existed during the twelve months post treatment. (Individuals in treatment 0 could have obtained one of these accounts if they learned about their existence independent of our marketing team and could have obtained the account opening documents by calling the partner bank's toll-free number in the United States.)

The coefficients in the first column of regressions a on treatments 1, 2, and 3 are all positive in sign and are each statistically significantly different from 0 at the 1 percent level. The coefficients indicate that treatments 1, 2, and 3 led, respectively, to 15.0, 14.1, and 22.1 percent points higher likelihood of owning a Cuenta Unidos account.

²⁹ These "other" accounts would have already existed prior to treatment or, if new, would have been opened without the assistance of our project staff.

²⁵ It is also the case that among observations assigned to treatments 2 and 3, attrition from the follow-up survey is statistically significantly lower for those with higher post treatment savings at the partner bank (as observed in the partner bank's internal data).

²⁶ We also implemented follow-up surveys of households of the primary remittance recipient in El Salvador. This survey suffered from even higher attrition and similar problems with baseline imbalance in partner bank savings. We do not present here treatment impacts on outcomes measured in these El Salvador household surveys. That said, impacts on savings in this sample are consistent with those found in the migrant follow-up sample.

 $^{^{27}}$ The indicator is equal to 1 if such an account exists at any time during months 1 to 12 posttreatment (including accounts that may have been open for only part of this period) and 0 otherwise. 28 Pure transferred to the second second

²⁸ Due to restrictions on how the partner bank was willing to share data with us, we cannot actually differentiate in the partner bank administrative data between Cuenta Unidos accounts held by both migrants and someone in El Salvador and Cuenta Unidos accounts held by only someone in El Salvador (without migrant co-ownership). However, we know from our project marketing records that in treatments 2 and 3, not a single migrant who opened an account for someone in El Salvador in treatments 2 or 3 opted to forgo joint ownership of the account. In treatment 1, all accounts opened with the assistance of our project staff were in the name of a person in El Salvador alone (consistent with instructions for that treatment). In all treatments, migrants could have found other ways of opening Cuenta Unidos accounts without our assistance, and if they did so, the accounts would be either joint accounts with the migrant or accounts in the name of an El Salvador person alone.

	Cuenta Unid (in name o in El Sa (a	los Accounts f someone lvador) ^a a)	Ahorro Accounts of migra (b	Directo (in name ant only)	Other Accou of Migrant Remittance	ints in Name or Primary e Recipient c)
Treatment 3 (joint account + migrant-only account)	0.221***	0.204***	0.234***	0.238***	0.020	0.014
	(0.034)	(0.034)	(0.033)	(0.031)	(0.042)	(0.041)
Treatment 2 (joint account)	0.141***	0.125***	-0.025	-0.018	-0.003	0.006
	(0.030)	(0.031)	(0.016)	(0.017)	(0.041)	(0.040)
Treatment 1 (PRR account only)	0.150***	0.135***	-0.013	-0.006	-0.027	-0.028
•	(0.032)	(0.032)	(0.018)	(0.018)	(0.042)	(0.040)
Constant	0.059***	0.557***	0.041***	0.250	0.265***	0.715***
	(0.016)	(0.191)	(0.013)	(0.164)	(0.030)	(0.243)
Control variables	. ,	Yes	. ,	Yes	. ,	Yes
Observations	898	898	898	898	898	898
R^2	0.041	0.145	0.141	0.217	0.001	0.078
<i>P</i> -value of <i>F</i> -test: equality of						
Treatment 3 and 2 coefficients	0.046	0.040	0.000	0.000	0.591	0.858
Treatment 3 and 1 coefficients	0.088	0.081	0.000	0.000	0.274	0.316
Treatment 2 and 1 coefficients	0.819	0.784	0.391	0.450	0.555	0.401

TABLE 2.—IMPACT OF TREATMENTS ON ACCOUNT OWNERSHIP AT PARTNER BANK (ORDINARY LEAST-SQUARES ESTIMATES) DEPENDENT VARIABLE: INDICATOR FOR EXISTENCE OF GIVEN TYPE OF ACCOUNT AT PARTNER BANK DURING TWELVE MONTHS POST-TREATMENT

Robust standard errors in parentheses. Significant at *10%, **5%, and ***1%. Dependent variable equal to 1 if migrant or remittance recipient has given type of project account with partner bank (Banco Agricola); 0 otherwise. Omitted treatment indicator is for treatment 0 (comparison group). Control variables: marketer fixed effects for the specific individual (out of nine) who conducted the marketing visit; stratification cell fixed effects for each of 48 unique combinations of stratification variables: ender (male/female). having a US, bank account (ves/no), relationship to remittance recipient (hid/source)(the), and

tion cell fixed effects for each of 48 unique combinations of stratification variables: gender (male/female), having a U.S. bank account (yes/no), relationship to remittance recipient (parent/child/spouse/other), and years in United States category (0-5 years/6-10 years/1-15 years); treatment month fixed effects; indicator for migrant demand for control. Treatment months are November 2007 through July 2008 inclusive. ^a Cuenta Unidos accounts opened with project assistance in treatments 2 and 3 are all joint accounts shared by migrants and someone in El Salvador, while those opened in treatment 1 are all accounts in the name of the PRR only. Some Cuenta Unidos accounts may have been opened without project assistance in any of the treatment groups, and in these cases, the accounts may be in the name of PRRs alone or joint between migrants and someone in El Salvador.

These coefficients are very similar when control variables were included in the regression.

Regressions (b) of table 2 are similar, except that the dependent variable is an indicator for the existence of an Ahorro Directo (migrant-only) account. The constant term in the first column of regressions b indicates that 4.1 percent of migrants in the comparison group opened such accounts (independent of the assistance of our project). The proportion is similar among migrants in treatments 1 and 2: the coefficients on the indicators for those treatments are small and not statistically different from 0. The coefficient on treatment 3, however, is large and statistically significant at the 1 percent level, indicating that migrants in that treatment condition were 23 to 24 percent points more likely to open an Ahorro Directo account than those in the comparison group.

Finally, regressions (c) replace the dependent variable with an indicator for the migrant owning some other account at the partner bank. The coefficient on the constant in the first column of (c) indicates that in 26.5 percent of migrant-PRR pairs, either migrant or PRR has another account at the partner bank. It appears that none of the treatments led to increased ownership of these other accounts; there is no large or statistically significant effect of any of the treatments on this outcome variable in either regression for this dependent variable.

The bottom rows of table 2 present *p*-values of *F*-tests of the difference between pairs of treatment coefficients. For opening of Cuenta Unidos accounts, the impact of treatment 3 is statistically significantly different from the impact of treatment 2 at the 5 percent level and from the impact of treatment 1 at the 10 percent level. This pattern is also exhibited (at 1 percent significance levels) in regressions (b) (for opening of Ahorro Directo accounts). The impact of

treatment 2 is not statistically significantly different from the impact of treatment 1 at conventional significance levels in any of the regressions in the table.

B. Impact on Savings at the Partner Bank

We estimate equation (1) to examine the impact of the treatments on savings balances at the partner bank. In table 3, the dependent variables are average savings balances over the twelve months after treatment for the various categories of accounts. Regressions where dependent variables are expressed as the quartic root of savings are presented in panel A of each table, with corresponding results for savings in dollars in panel B.

In the first two columns of table 3, the dependent variable is savings in Cuenta Unidos accounts.³⁰ The first column reports coefficient estimates for regressions without control variables, while the second column provides corresponding estimates but where control variables are included in the regression (this format is repeated for other dependent variables in subsequent columns). All treatments have positive impacts on Cuenta Unidos savings balances. Estimates in panel A indicate that each treatment has a positive effect on the quartic root of savings, all at conventional levels of statistical significance. In panel B, coefficient estimates of the impact on the dollar value of savings are also positive but

³⁰ As mentioned above, due to the ambiguity in the partner bank's database, we cannot distinguish between Cuenta Unidos savings in joint accounts from Cuenta Unidos savings in accounts in the name of someone in El Salvador alone. However, due to the assistance we provided in account opening in treatments 1, 2, and 3, it is most likely that in treatments 2 and 3, accounts opened in this project are joint accounts, while in treatment 1, they are most likely not joint accounts. In treatment 0, the few observed Cuenta Unidos accounts were opened without our staff's assistance, so we do not know whether these are joint accounts or not.

DEPENDENT V ARL	ABLE: JAVINGS DAL	ANCE (US\$), AVERP	GE OVER I WELVE I	VIONTHS FOSTIREA	IMENT, IN ACCOUNTS	OF CIVEN 1 YPE		
	Cuenta Unio (in name o	dos Accounts of someone	Ahorro Accounts	Directo (in name	Other Accou of migrant	nts (in name or primary	In Total	Across
	in El Se	alvador) ^a a)	of migr	ant only)	remittance	recipient)	All Ac	counts $(h) \pm (c)$
		(a)		(n			$(\mathbf{n}) = (\mathbf{a})$	$(n) \pm (n)$
A: Quartic Root	***T000 0	0 051***	*** 100 0	*** 900 0	695.0	2000	*** 200 0	×**007 0
1 reaument 3 (joint account + migrant-only account)	0.38/	0.334 **** (0 114)	0.281	(1000 0)	0.282	0.200	0.715)	0.039
Treatment 2 (joint account)	0.232**	0.231^{**}	-0.079**	-0.057	0.035	-0.006	0.110	0.102
~	(0.101)	(0.109)	(0.039)	(0.047)	(0.177)	(0.178)	(0.193)	(0.198)
Treatment 1 (PRR account only)	0.184^{**}	0.162*	-0.012	-0.004	-0.084	-0.122	0.044	0.001
	(0.092)	(0.088)	(0.051)	(0.051)	(0.179)	(0.180)	(0.193)	(0.195)
Constant	0.176***	0.920	0.079**	0.004	0.887***	2.747**	1.119***	2.818**
Control worklas	(ccn.n)	(0.028) V ₂₂	(660.0)	(161.0) V25	(0.124)	(681.1) V25	(661.0)	(1.184)
Collutor Variables	606	105	609	1 CS	606	1 55	609	1CS 806
R^2	0.013	0.087	0.037	0.117	0.004	0.064	0.016	0.082
<i>p</i> -value of <i>F</i> -test: Equality of								
Treatment 3 and 2 coefficients	0.214	0.334	0.000	0.000	0.220	0.305	0.008	0.017
Treatment 3 and 1 coefficients	0.088	0.101	0.001	0.000	0.072	0.105	0.003	0.004
Treatment 2 and 1 coefficients	0.671	0.555	0.048	0.091	0.514	0.531	0.741	0.624
Mean of dependent variable in comparison group		0.176		0.079		0.887		1.119
B: In Dollars								
Treatment 3 (joint account + migrant-only account)	79.770	106.543	28.978*	32.533*	187.826*	142.479	296.574**	281.555**
	(68.482)	(83.989)	(15.166)	(17.131)	(106.129)	(101.002)	(126.380)	(132.572)
Treatment 2 (joint account)	70.062	94.661^{*}	-9.074	-6.996	86.995	25.354	147.983	113.018
	(45.054)	(55.237)	(6.920)	(8.159)	(109.439)	(110.911)	(119.543)	(126.260)
Treatment 1 (PRR account only)	12.693	27.980	-5.605	-5.802	38.339	1.988	45.426	24.166
	(13.961)	(21.121)	(7.211)	(8.338)	(101.573)	(102.704)	(102.511)	(105.030)
Constant	16.005*	240.552	9.074	6.610	160.836^{***}	382.230	185.914^{***}	629.391^{*}
	(8.469)	(163.721)	(6.920)	(15.665)	(56.859)	(317.066)	(57.586)	(366.516)
Control variables		Yes		Yes		Yes		Yes
Observations	898	898	898	898	898	898	898	898
R^2	0.003	0.035	0.017	0.087	0.003	0.069	0.006	0.056
p-value of F -test: Equality of								
Treatment 3 and 2 coefficients	0.905	0.893	0.005	0.008	0.435	0.357	0.332	0.285
Treatment 3 and 1 coefficients	0.329	0.280	0.012	0.015	0.223	0.243	0.074	0.069
Treatment 2 and 1 coefficients	0.208	0.197	0.094	0.713	0.700	0.861	0.448	0.540
Mean of dependent variable in comparison group		16.005		9.074		160.836		185.914
Robust standard errors in parentheses. Significant at *10%, **5%, and ***1% * Cuenta Unidos accounts opened with project assistance in treatments 2 and	%. Dependent variables a 13 are all joint accounts s	re averaged over end-of-i hared by migrants and so	nonth balances in U.S. meone in El Salvador,	dollars. See table 2 for on while those opened in the	other notes. eatment 1 are all accounts	in the name of the PRI	R only. Some Cuenta Unid	os accounts may have been
opened without project assistance in any of the treatment groups, and in these ca	ases, the accounts may b	e in the name of PRRs al	one or joint between mi	grants and someone in I	El Salvador.			

TABLE 3.—IMPACT OF TREATMENTS ON SAVINGS IN ACCOUNTS AT PARTNER BANK (ORDINARY LEAST-SQUARES ESTIMATES)

342

THE REVIEW OF ECONOMICS AND STATISTICS

are mostly not statistically significantly different from zero at conventional levels.

Results in the third and fourth columns of table 3 reveal positive impacts of treatment 3 on savings in Ahorro Directo accounts. These effects are statistically significant at the 1 percent level for the quartic root of savings and at the 10 percent level for savings in dollars.³¹

In the fifth and sixth columns, where the dependent variables refer to savings in other partner bank accounts, point estimates of the effects of treatment 3 are large in magnitude, but with one exception (in the first regression in panel B), they are not statistically significantly different from 0. Estimated effects of treatments 2 and 1 are mostly small in magnitude, inconsistent in sign across specifications, and not statistically significantly different from 0.

In the last two columns of the table, dependent variables refer to total savings at the partner bank. Impacts of treatment 3 are all large, positive, and statistically significantly different from 0 (at the 1 percent level in panel A and the 5 percent level in panel B). Point estimates of the effects of treatments 2 and 1 are all also positive but are all smaller in magnitude and are not statistically significantly different from 0. In panel A, we can reject that the effect of treatment 3 is equal to the effect of either treatments 2 or 1 at conventional significance levels (5 percent or better). In panel B, the effect of treatment 3 cannot be distinguished from that of treatment 2 at conventional levels, but is statistically significantly larger than the effect of treatment 1 at the 10 percent level.

The result in the last column (with control variables) of panel B indicates that treatment 3 led total savings at the partner bank to be larger by \$282. This is a large increase over mean partner bank savings in the comparison group (\$186).

To provide a sense of the percentiles of the savings distribution that are contributing to these treatment effects, figure 1 presents the cumulative distribution function (CDF) of the quartic root of total savings in all project accounts (the dependent variable of the last two columns of table 3, panel A). The CDF is truncated on the vertical axis at the 50th percentile to enhance visibility.³² The CDF for treatment 3 is clearly shifted to the right compared to the CDFs of the other treatments. While treatment effects show up relatively high in the savings distribution, it is far from the case that the estimated impacts of the savings distribution.

³² Recall from table 2 that at most (in treatment 3), only 55 percent of observations had any account at the partner bank, so it is expected that there are many zeros in the data. The percent of observations with zero savings at the partner bank in treatments 3, 2, 1, and 0 are, respectively, 55 percent, 66 percent, 64 percent, and 68 percent.





CDF truncated at 0.5 on the vertical axis. The variable depicted is identical to the dependent variable in regressions of table 3, panel A, column d.

C. Exploring Shifting across Savings Mechanisms

A question that naturally arises at this point is whether the treatments led to increases in savings overall or whether increases in savings seen at the partner bank were simply shifted from other savings mechanisms (e.g., cash or other banks). That said, even if all increases we find in partner bank savings were simply reallocations of existing savings held elsewhere, our results so far would still be revealing of migrant demand for control over home country savings. Reallocation of savings across savings mechanisms is in itself a consequential financial behavior that individuals are not likely to take lightly.

The most natural type of savings reallocation to examine, which we turn to first, is shifting of funds from other partner bank accounts to the project accounts. This is an important type of shifting to examine, since if study participants do shift, it should be easiest to do so from one account to another within the partner bank. The results in table 3 allow us to easily check for evidence of shifting of funds within the partner bank. A pattern that would be consistent with shifting would be negative coefficients on treatment coefficients in columns (c), where the dependent variable is savings in other (non-project) accounts. In addition, if such shifting was large enough, we could find no statistically significant impact on total savings at the partner bank (columns d).

It turns out that neither pattern emerges. Coefficients in both column (c) regressions are mostly positive (across panels A and B), and when they are negative, they are small in magnitude and not statistically significantly different from 0. Indeed, the coefficients on treatment 3 in column (c) in both panels are positive, large in magnitude, and hover around the threshold of statistical significance, indicating, if anything, a positive spillover toward "other" sav-

³¹ Interestingly, coefficients are actually negative for treatments 2 and 1 in these regressions. This may reflect the fact that marketing treatments 2 and 1 focused on encouraging savings in accounts of remittance recipients and did not encourage migrant-only accounts. However, we do not highlight these results, since these impacts are mostly not statistically significantly different from 0 (and the statistically significant treatment 2 coefficient in column 3 is not robust to inclusion of control variables).

	Savings in El Salvador (a)	Savings in the United States (b)	Savings in Cash (c)	In Total (d) = (a) + (b) + (c)
A: Ouartic Root				
Treatment 3 (joint account + migrant-only account)	0.366	0.711*	-0.445 **	0.516
	(0.302)	(0.387)	(0.197)	(0.436)
Treatment 2 (joint account)	0.463	0.227	-0.080	0.581
•	(0.316)	(0.360)	(0.202)	(0.436)
Treatment 1 (PRR account only)	0.239	0.173	-0.066	0.266
•	(0.293)	(0.362)	(0.232)	(0.433)
Control variables	Yes	Yes	Yes	Yes
Observations	508	508	508	508
R^2	0.127	0.167	0.147	0.149
<i>p</i> -value of <i>F</i> -test: equality of				
Treatment 3 and 2 coefficients	0.734	0.186	0.048	0.874
Treatment 3 and 1 coefficients	0.636	0.156	0.079	0.556
Treatment 2 and 1 coefficients	0.442	0.873	0.949	0.443
Mean of dependent variable in comparison group	0.629	1.573	0.552	2.567
B: In Dollars				
Treatment 3 (joint account $+$ migrant-only account)	382.002	450.125	-188.496**	643.630
	(271.230)	(356.568)	(94.733)	(455.407)
Treatment 2 (joint account)	607.438*	-181.102	-19.996	406.340
° ,	(342.681)	(305.878)	(100.273)	(468.693)
Treatment 1 (PRR account only)	193.795	-122.822	84.552	155.526
· · · · ·	(242.484)	(297.461)	(127.252)	(403.441)
Control variables	Yes	Yes	Yes	Yes
Observations	508	508	508	508
R^2	0.087	0.142	0.102	0.119
<i>p</i> -value of <i>F</i> -test: equality of				
Treatment 3 and 2 coefficients	0.457	0.030	0.031	0.577
Treatment 3 and 1 coefficients	0.425	0.064	0.013	0.220
Treatment 2 and 1 coefficients	0.189	0.747	0.338	0.494
Mean of dependent variable in comparison group	264.732	599.866	143.036	1007.634

TABLE 4.—IMPACT OF TREATMENTS ON SAVINGS REPORTED BY MIGRANTS IN FOLLOW-UP SURVEY (ORDINARY LEAST-SQUARES ESTIMATES) DEPENDENT VARIABLE: SAVINGS REPORTED IN FOLLOW-UP SURVEY (MARCH–JUNE 2009)

Robust standard errors in parentheses. Significant at *10%, **5%, and ***1%. Follow-up survey administered from March–June 2009. Dependent variable is stock of savings in U.S. dollars. Omitted treatment indicator is for treatment 0 (comparison group). Control variables: marketer fixed effects are for the specific individual (out of nine) who conducted the marketing visit; stratification cell fixed effects for each of 48 unique combinations of stratification variables: gender (male/female), having a U.S. bank account (yes/no), relationship to remittance recipient (parent/child/spouse/other), and years in US category (0–5 years/6–10 years/1–15 years); treatment month fixed effects; indicator for migrant demand for control. Treatment months are November 2007 through July 2008 inclusive.

ing (rather than shifting). In addition, column (d) indicates that overall, total savings at the partner bank did increase in response to treatment 3, an effect that is statistically significant at conventional levels in both panels. The results in table 3, in sum, provide no evidence that treatment effects are driven by shifting of funds within the partner bank.

Of course, shifting could also occur from funds held outside the partner bank. It is therefore also useful to examine impacts of the treatments on the stock of savings more broadly, as reported by respondents in the follow-up survey (administered roughly one year after the intervention). Negative impacts on certain types of savings would be revealing of shifting, while there is of course the possibility of positive spillovers on other types of savings, perhaps due to the marketing pitch delivered with the new account offers.³³

Table 4 presents regression estimates of the impact of each treatment on savings reported by migrants (the dependent variables are expressed as the quartic root of savings in panel A and in dollars in panel B). The four columns present impacts on savings reported by the D.C.-based migrant (a) in banks in El Salvador, (b) in banks in the United States, (c) in cash, and (d) in total across the previous three categories. In both panels, effects of treatment 3 are positive and large in magnitude for savings in El Salvador banks, in U.S. banks, and in total, but most of these coefficients are not statistically significantly different from 0 at conventional levels. The exception is the impact of treatment 3 on the quartic root of savings in the United States (panel A, column b) which is statistically significantly different from 0 at the 10 percent level. It also appears that the treatment shifted savings away from cash: the treatment 3 coefficient in column c is negative and significant at the 5 percent level in both panels A and B.

It appears that treatment 3 had a positive and statistically significant impact on migrant savings in the United States. We note that the point estimates of impacts of treatment 3 on the dollar value of El Salvador bank savings and on total savings (panel B of table 4) are positive and larger in magnitude than the impacts on total savings at the partner bank

³³ The qualitative results of table 3 (from regressions with the full 898observation sample) carry through in the smaller (N = 508) follow-up survey sample. Impacts of the treatments on partner bank savings in this subsample are presented in online appendix table 4. The pattern of impacts on partner bank savings in the twelve months post treatment (as well as statistical significance levels) in the smaller follow-up samples corresponds to those in the full sample, and in many cases point estimates are even larger in magnitude.

(see online appendix table 4). If the impacts on total savings at the partner bank were simply due to shifting of funds from other savings vehicles, then treatment effects in table 4 should have been smaller in magnitude than those in online appendix table 4. There is therefore no indication that the impacts on total savings at the partner bank are simply due to shifting from other savings vehicles.

However, recall that (as discussed in section III) this follow-up sample suffers from differential attrition related to treatment, as well as imbalance in baseline pretreatment savings at the partner bank across treatment conditions. These estimates from the follow-up survey data should therefore be interpreted with caution because they may reflect selection bias.³⁴

V. Discussion and Additional Analyses

We now present additional analyses and discussion to help interpret our results, explore impacts beyond the first twelve months, and examine impacts on remittances.

A. Heterogeneity in Treatment Effects by Baseline Demand for Control

A central motivation of this study is the hypothesis that the accounts offered to migrants (particularly those in treatments 2 and 3) would lead to increased savings because they allowed migrants to exert greater control over savings in the home country. To test this interpretation of our results, we examine the extent to which our treatment effects are heterogeneous vis-à-vis the extent to which migrants, in the baseline survey, expressed demand for control over financial decision making in El Salvador primary remittance recipient households.

Estimates of treatment effect heterogeneity are presented in table 5. The regressions are analogous to those of table 3 where the dependent variables are different types of savings at the partner bank, but now the treatment indicators are each interacted with an indicator for the migrant expressing demand for control at baseline (defined in section III; 51 percent of migrants have "demand for control"), and an indicator for "no demand for control" (simply 1 minus the demand for control indicator). The regressions include all control variables (including the main effect of the "demand for control" indicator).³⁵

The coefficient on each interaction term is the effect of the given treatment on savings for migrants with or without baseline demand for control. *F*-tests (with *p*-values reported at the bottom of each panel) test the null that the treatment effect is the same across migrants with versus without demand for control.

In column (a) of panel A, the dependent variable is the quartic root of savings in Cuenta Unidos accounts. The coefficient on Treatment $3 \times$ Demand for Control is positive and significant at the 1 percent level, while that on Treatment $3 \times$ No Demand for Control is much smaller in magnitude and is not significant at conventional levels. This pattern also holds for the treatment 2 interaction terms. In both cases, an F-test rejects (at the 5 percent level) equality of the treatment effects across migrants with and without demand for control. In column (a) of panel B, where the dependent variable is expressed in dollars, the qualitative pattern of results is the same, but significance levels are lower and F-tests cannot reject the equality of coefficients across migrants with and without demand for control. This pattern of treatment effect heterogeneity for treatments 3 and 2 is consistent with migrants exerting control over remittance-recipient savings in accounts jointly held by migrants and remittance recipients.

Interestingly, unlike in column (a), in column (b) (where the dependent variable is savings in migrant-only Ahorro Directo accounts), it is not the case that treatment 3 has a greater impact on savings among migrants with demand for control. In fact, the pattern is reversed: in both panels, only Treatment $3 \times$ (No Demand for Control) is statistically significant at conventional levels; it is substantially larger in magnitude than the coefficient on Treatment $3 \times$ (Demand for Control). *F*-tests reject (at the 5 percent level) the equality of the coefficients.

The contrasting results in columns (a) and (b) suggest that migrant desire to control savings decision making is associated with the objectives behind savings in the home country, and therefore choice of savings product. Migrants with demand for control may desire savings to be used primarily for objectives of the remittance recipient and seek to exert control to help ensure the objectives are met. For example, migrants may seek control over savings to help El Salvador households build up buffer stocks (precautionary savings) that can be accessed quickly by the household in emergencies. This may be the reason behind the differentially large effect of treatment 3 on joint account savings (columns a and b) for migrants with demand for control.

Migrants without demand for control may be saving primarily to achieve their own objectives (objectives not shared with the remittance recipient). For these migrants, access to bank accounts by El Salvador individuals is immaterial, so they are not attracted to joint accounts. For example, such migrants may simply want to keep some savings in El Salvador for themselves, to provide easy access during visits home or as a safe place to keep funds in case the migrant is deported and faces difficulty accessing U.S. bank accounts. These migrants therefore save only in the migrant-only accounts. This may explain the differentially large effect of treatment 3 for migrants without demand for control in column (b).

In columns (c) and (d), analogous results are presented for savings in other types of accounts (not offered by the project) and for total savings at the partner bank. In panel A's results for the quartic root of these savings variables,

³⁴ That said, including controls in the regression for baseline savings at the partner bank (the quartic root and in dollars) has essentially no effect on the treatment coefficient point estimates or standard errors in table 4.

³⁵ Main effects for each treatment do not need to be included because they are fully interacted with "demand for control" and "no demand for control."

	Cuenta Unidos Accounts (in name of someone in El Salvador)	Ahorro Directo Accounts (in name of migrant only)	Other Accounts (in name of migrant or primary remittance recipient)	In Total Across All Accounts
	(a)	(b)	(c)	(d) = (a) + (b) + (c)
A: Ouartic Root				
Treatment $3 \times$ Demand for Control	0.572***	0.113	0.331	0.768***
	(0.171)	(0.110)	(0.261)	(0.293)
Treatment 3 \times No Demand for Control	0.117	0.511***	0.070	0.497*
	(0.135)	(0.146)	(0.279)	(0.302)
Treatment 2 \times Demand for Control	0.478***	-0.119*	0.047	0.283
	(0.161)	(0.072)	(0.254)	(0.286)
Treatment 2 \times No Demand for Control	-0.022	0.009	-0.054	-0.074
	(0.134)	(0.044)	(0.256)	(0.275)
Treatment $1 \times Demand$ for Control	0.121	-0.114	-0.355	-0.340
	(0.110)	(0.076)	(0.231)	(0.253)
Treatment $1 \times No$ Demand for Control	0.186	0.114*	0.099	0.326
	(0.140)	(0.066)	(0.277)	(0.298)
Control variables	Yes	Yes	Yes	Yes
Observations	898	898	898	898
R^2	0.098	0.126	0.068	0.090
<i>p</i> -value of <i>F</i> -test: equality of interactions with				
Treatment 3	0.031	0.025	0.490	0.516
Treatment 2	0.014	0.100	0.783	0.372
Treatment 1	0.722	0.024	0.209	0.091
Mean of dependent variable in comparison group				
Migrants with demand for control	0.086	0.129	0.923	1.113
Migrants with no demand for control	0.277	0.022	0.85	1.13
B: In Dollars				
Treatment $3 \times$ Demand for Control	189.768	-4.669	299.785**	484.884**
	(148.826)	(14.858)	(130.469)	(199.087)
Treatment $3 \times No$ Demand for Control	16.499	72.243**	-27.013	61.729
	(27.879)	(29.198)	(148.808)	(153.673)
Treatment 2 \times Demand for Control	162.918*	-17.819	118.489	263.588*
	(98.902)	(12.954)	(111.959)	(155.977)
Treatment 2 \times No Demand for Control	24.894	4.335	-67.874	-38.645
	(35.275)	(6.938)	(195.811)	(198.076)
Treatment 1 \times Demand for Control	14.106	-18.008	-65.784	-69.687
	(22.881)	(14.363)	(96.316)	(98.699)
Treatment $1 \times No$ Demand for Control	36.222	7.967	59.510	103.699
	(30.408)	(6.471)	(168.831)	(171.613)
Control variables	Yes	Yes	Yes	Yes
Observations	898	898	898	898
R^2	0.039	0.101	0.074	0.062
<i>p</i> -value of <i>F</i> -test: equality of interactions with				
Treatment 3	0.212	0.013	0.089	0.077
Treatment 2	0.173	0.098	0.411	0.228
Treatment 1	0.541	0.086	0.494	0.352
Mean of dependent variable in comparison group				
Migrants with demand for control	13.637	16.907	94.118	124.662
Migrants with no demand for control	18.672	0.252	235.974	254.898

TABLE 5.—HETEROGENEITY IN TREATMENT EFFECTS BY BASELINE DEMAND FOR CONTROL (ORDINARY LEAST-SQUARES ESTIMATES) DEPENDENT VARIABLE: SAVINGS BALANCE (US\$), AVERAGE OVER TWELVE MONTHS POSTTREATMENT, IN ACCOUNTS OF GIVEN TYPE

Robust standard errors in parentheses. Significant at *10%, **5%, and ***1%. Dependent variables are averaged over end-of-month balances in U.S. dollars. Regressions need not include main effects of treatments 3, 2, and 1 because they are fully interacted with "demand for control" and "no demand for control." See table 2 for other notes.

while the coefficient on the treatment 3 interaction term is larger for migrants with than without demand for control, we cannot reject at conventional levels that these coefficients are equal to one another. In panel B, on the other hand (savings in dollars), the Treatment $3 \times$ Demand for Control coefficient is larger than that on Treatment $3 \times$ No Demand for Control at the 10 percent level in both columns.

B. Decoy Effects

One pattern in the results is that the impacts related to the joint account (Cuenta Unidos) are higher in treatment 3 than

in treatment 2. This pattern is most prominent in the analysis of impacts on account opening in table 2, regressions (a) the coefficient on treatment 3 is larger than that on treatment 2 (by about 8 percent points), and the difference between the two is statistically significantly different from 0 at the 5 percent level. The same qualitative pattern also arises for treatment effects on savings in joint accounts (first two columns of table 3, panel A): the coefficient on treatment 3 is larger than the coefficient on treatment 2, although differences are not statistically significant in this case.

This pattern is a bit of a puzzle for standard models of economic decision making, since the Cuenta Unidos joint account was offered in both treatments 3 and 2. One explanation for this result is that the offer of the migrant-only Ahorro Directo account in treatment 3 had a "decoy effect" on demand for the joint account.³⁶ Other research has documented decoy effects, or shifts in preference for a certain option when presented with another option that might be thought to be irrelevant. For example, Laran et al. (2011) find that priming brands such as Walmart increase consumers' cost consciousness and subsequently reduce their spending intentions, even on alternative products. Chatterjee and Rose (2012) found that people primed with cash would then focus on the costs of products, while those primed with credit cards would focus on the benefits. Then they introduced "decoy products": in the cash option, they introduced a decoy for the benefits choice, and in the credit card option, they introduced a decoy for the costs choice. This countered the initial priming, reducing the salience of the benefits choice for the credit card prime and of the cost choice for the cash prime. In other words, the existence of an additional product with certain features functions exactly as priming through other sources does: it focuses the attention of the individual on those features in evaluating all the products.

In our study, migrants who were offered both the joint account and the migrant-only account (treatment 3) were more likely to open the joint account than those who were offered only the joint account (treatment 2). That is, the presence of a third option changed the migrants' valuation of the joint account. We believe this likely represents a decoy effect in the sense of the literature referenced above. A related concept in economics is a menu effect; both are violations of the independence of irrelevant alternatives axiom.

In our case, we think the effect of the option of a private savings account focused the migrants on the control features of the joint account, increasing their valuation of the joint account. Marketing of the migrant-only savings product emphasized the importance of control over savings. It is likely that this caused migrants to consider the control aspects of the joint account more so than when the joint account was offered in isolation. As in Laran et al. (2011), where the addition of Walmart to a choice set primed subjects to weight cost more, offering the migrant-only account likely focused migrants on the control features of the joint account.

C. Ruling Out That Effects Are Due to Marketing Pitch Alone

One question that arises is whether treatment effects are due to the marketing pitch alone or whether it is crucial that the intervention offered the accounts and account-opening assistance at the partner bank. The concern is that the set of marketing pitches implemented might have been enough to encourage migrants to exert control over funds in joint accounts that already existed or that they could easily set up on their own. Then the intervention's offer of the joint accounts at the partner bank (and account-opening help) may have been superfluous.

To test this, we use the migrant follow-up survey data to check whether treatment 3 led to increases in savings held jointly by migrants and El Salvador individuals at other (non-partner) banks. If the intervention's offer of assistance opening joint accounts at the partner bank was superfluous and the marketing pitch was all that mattered, then we should also see treatment 3 have positive effects on savings at other banks (many of whose branch locations may have been more conveniently located for family members in El Salvador).

Regression results are in online appendix table 5. The dependent variables in the two columns are savings reported by the migrant in joint accounts outside the partner bank shared with primary remittance recipients (column 1) and with other people, not including the primary remittance recipient (column 2). Dependent variables are expressed as the quartic root in panel A and in dollars in panel B. There is no indication that treatment 3 affects savings in joint accounts with primary remittance recipients outside the partner bank: treatment 3 coefficients in columns 1 or 2 are statistically significantly different from 0. (Interestingly, there does seem to be an effect of treatment 2 on joint savings with non-PRRs outside the partner bank, which is positive and statistically significant at the 10 percent level.)

We conclude from this analysis that the marketing pitch alone cannot explain treatment 3's impact on savings at the partner bank and that it was crucial that the treatment offered actual bank accounts and assistance in opening them.

D. Longer-Term Impacts

Our results so far refer to savings roughly one year after treatment. Outcomes in administrative data are average savings balances over the twelve posttreatment months, while outcomes in the follow-up survey refer to savings stocks about twelve months post treatment. An important question is whether the effects of the treatment extended further in time. Because we did not administer further follow-up surveys, this analysis is limited to savings outcomes in the administrative data of our partner bank.

It is important to note that our agreement with the partner bank was that general marketing of the new products designed for the project (Cuenta Unidos and Ahorro Directo) would be restricted until administration of the follow-up survey. During this period, the new products were offered only by our project marketing staff on a face-to-face basis to study participants; there was no marketing of these products to customers more generally. This was done to reduce the extent to which marketing spillovers across treat-

³⁶ Recall that the pitch for the joint account was the same in both treatments 2 and 3. The difference between the treatments was that treatment 3 also offered the Ahorro Directo migrant-only account, with its own separate pitch.

THE REVIEW OF ECONOMICS AND STATISTICS

Posttreatment months:	1–6	7–12	13–18	19–24	25-30	31–36	37-42	43-48
A: Quartic Root								
Treatment 3 (joint account + individual migrant account)	0.634***	0.626***	0.475**	0.326	0.275	0.322	0.255	0.212
ч с <i>,</i>	(0.207)	(0.214)	(0.221)	(0.206)	(0.207)	(0.210)	(0.207)	(0.218)
Treatment 2 (joint account)	0.143	0.091	-0.047	-0.045	-0.091	-0.119	-0.122	-0.125
•	(0.195)	(0.197)	(0.196)	(0.193)	(0.197)	(0.201)	(0.199)	(0.206)
Treatment 1 (remittance recipient account)	0.056	-0.058	-0.137	-0.172	-0.150	-0.131	-0.096	-0.204
	(0.190)	(0.192)	(0.200)	(0.203)	(0.212)	(0.223)	(0.220)	(0.221)
Control variables	Yes							
Observations	898	898	898	898	898	898	898	898
R^2	0.083	0.082	0.074	0.067	0.064	0.068	0.070	0.070
<i>p</i> -value of <i>F</i> -test: equality of								
Treatment 3 and 2 coefficients	0.028	0.016	0.017	0.075	0.075	0.033	0.074	0.127
Treatment 3 and 1 coefficients	0.007	0.002	0.006	0.024	0.061	0.052	0.132	0.079
Treatment 2 and 1 coefficients	0.671	0.463	0.655	0.544	0.787	0.958	0.903	0.714
Mean of dependent variable in comparison group	1.051	1.056	1.124	1.122	1.112	1.112	1.083	1.152
B: In Dollars								
Treatment 3 (joint account + individual migrant account)	277.787**	285.323*	211.905	213.830	153.626	31.484	20.687	31.584
	(120.166)	(156.322)	(227.437)	(219.450)	(211.533)	(169.174)	(220.522)	(274.700)
Treatment 2 (joint account)	152.104	73.933	-32.024	37.460	25.950	-35.547	-64.035	-111.047
-	(122.947)	(135.507)	(122.470)	(129.029)	(170.891)	(201.742)	(215.868)	(238.924)
Treatment 1 (remittance recipient account)	40.049	8.284	7.155	107.273	203.372	178.405	142.401	56.479
	(96.619)	(119.421)	(126.208)	(127.078)	(183.632)	(215.878)	(210.993)	(228.290)
Control variables	Yes							
Observations	898	898	898	898	898	898	898	898
R^2	0.061	0.051	0.033	0.034	0.036	0.048	0.045	0.046
<i>p</i> -value of <i>F</i> -test: equality of:								
Treatment 3 and 2 coefficients	0.422	0.207	0.234	0.401	0.546	0.674	0.692	0.591
Treatment 3 and 1 coefficients	0.079	0.077	0.302	0.603	0.833	0.471	0.572	0.922
Treatment 2 and 1 coefficients	0.433	0.663	0.741	0.622	0.432	0.366	0.338	0.443
Mean of dependent variable in comparison group	165.620	206.209	288.509	230.890	248.745	301.652	304.683	357.046

TABLE 6.—IMPACT OF TREATMENTS ON SAVINGS AT PARTNER BANK, OVER TIME (ORDINARY LEAST-SQUARES ESTIMATES) DEPENDENT VARIABLE: SAVINGS BALANCE (US\$), AVERAGE OVER GIVEN MONTHS POSTTREATMENT, IN TOTAL ACROSS ALL PARTNER BANK ACCOUNTS

Robust standard errors in parentheses. Significant at *10%, **5%, and ***1%. Dependent variables are averaged over end-of-month balances in U.S. dollars. Omitted treatment indicator is for treatment 0 (comparison group). Control variables: marketer fixed effects are for the specific individual (out of nine) who conducted the marketing visit; stratification cell fixed effects for each of 48 unique combinations of stratification variables: gender (male/female), having a U.S. bank account (yes/no), relationship to remittance recipient (parent/child/spouse/other), and years in US category (0–5 years/6–10 years/11–15 years); treatment month fixed effects; indicator for migrant demand for control. Treatment months are November 2007 through July 2008, inclusive.

ment and comparison groups would contaminate (and attenuate) our treatment effect estimates. After the followup survey, roughly one year after the treatments were administered, the partner bank did begin marketing the Cuenta Unidos and Ahorro Directo accounts broadly in both the United States and El Salvador. Generalized take-up of the new products could therefore lead all treatment groups to become increasingly similar to one another in terms of product use, preventing the data from revealing whether treatment effects persist over time.

This does turn out to be the case. In table 6 we examine impacts of the treatments on total savings across accounts at the partner bank in 6-month windows up to 48 months posttreatment. (In all respects, the regressions are analogous to that of the last column of table 3, except that average savings balances are measured over differing months post treatment.) Results in the first two columns reflect results reported previously in table 3: there are positive and statistically significant effects of treatment 3 on total partner bank savings in months 1 to 6 and months 7 to 12 post treatment in both the quartic root and dollar specifications. The impact persists into months 13 to 18 post treatment: the coefficient on treatment 3 in the quartic specification remains positive and statistically significant at the 5 percent level (although the coefficient in the dollar specification has declined somewhat in magnitude and is no longer statistically significantly different from 0). In the remaining columns of the table, coefficients on treatment 3 in both panels decline further in magnitude, and in no case are they statistically significantly different from 0 at conventional levels.³⁷

These results do not allow us to tell whether the impact of treatment 3 is truly only temporary (lasting no more than 18 months posttreatment), or whether persistent treatment

³⁷ Online appendix figures 1 to 4 provide month-by-month detail on savings balances by treatment group. Appendix figure 1 displays monthly total savings balances, graphically depicting the pattern found in table 6: balances in treatment 3 are clearly higher through roughly months 12 to 18, after which balances in other treatment groups catch up. Appendix figures 2 and 3 display balances in, respectively, joint (Cuenta Unidos) and migrant-only (Ahorro Directo) accounts. For each of these types of accounts offered by the project, it is clear that after roughly the twelfth to eighteenth month post treatment, balances rise in the control group and other treatment groups that were not originally offered these accounts by our research project. When it comes to savings balances in other accounts at the partner bank (shown in appendix figure 4), the pattern is slightly different, with treatments 1 and 2 catching up with treatment 3 but with balances in the control group remaining persistently lower. Overall it appears that catch-up in the control group after months 12 to 18 is driven by control group savings in the project accounts (Cuenta Unidos and Ahorro Directo).

SAVINGS IN TRANSNATIONAL HOUSEHOLDS

Remittance recipient:	Anyone in El Salvador	Anyone in El Salvador	Primary Remittance Recipient
Remittance channel:	Partner Bank	Partner Bank	All Channels
Time Frame:	July 2008–June 2009	July 2008–June 2009	July 2008 –Follow-Up Survey
Sample:	Full Sample	Migrants Completing Follow-Up Survey	Migrants Completing Follow-Up Survey
Data Source:	Partner Bank Database	Partner Bank Database	Follow-Up Survey
A: Main Effect of Treatments			
Treatment 3 (joint account + migrant-only account)	10.659	18.132	35.940
	(18.778)	(24.477)	(52.405)
Treatment 2 (joint account)	-20.180	-9.358	-2.078
	(16.061)	(19.391)	(33.161)
Treatment 1 (PRR account only)	-24.121	-31.648	5.365
	(16.270)	(20.926)	(37.339)
Control variables	Yes	Yes	Yes
Observations	898	560	560
R^2	0.149	0.199	0.092
<i>p</i> -value of <i>F</i> -test: equality of			
Treatment 3 and 2 coefficients	0.081	0.236	0.391
Treatment 3 and 1 coefficients	0.053	0.035	0.528
Treatment 2 and 1 coefficients	0.781	0.261	0.810
Mean of dependent variable in comparison group	71.283	82.423	239.954
B: Separate Treatment Effects for Migrants with and wa	ithout Baseline Demand for Co	ontrol	
Treatment $3 \times$ Demand for Control	-5.036	-8.904	-10.517
	(29.416)	(33.463)	(53.487)
Treatment $3 \times No$ Demand for Control	27.516	47.373	88.036
	(20.963)	(31.321)	(86.988)
Treatment 2 \times Demand for Control	-30.120	-2.136	-15.025
	(27.596)	(32.783)	(57.674)
Treatment 2 \times No demand for Control	-9.584	-15.412	12.517
	(16.446)	(25.576)	(38.312)
Treatment $1 \times Demand$ for Control	-36.302	-59.460*	-2.512
	(27.427)	(31.719)	(61.83)
Treatment 1 \times No Demand for Control	-11.237	-1.655	14.382
	(19.254)	(31.352)	(36.958)
Control variables	Yes	Yes	Yes
Observations	898	560	560
R^2	0.150	0.204	0.094
<i>p</i> -value of <i>F</i> -test: equality of interactions with			
Treatment 3	0.355	0.193	0.310
Treatment 2	0.529	0.765	0.703
Treatment 1	0.469	0.222	0.812
Mean of dependent variable in comparison group			
Migrants with demand for control	91.473	99.412	277.754
Migrants with no demand for control	48.545	62.213	194.986

TABLE 7.—IMPACT OF TREATMENTS ON REMITTANCES SENT INTO BANK ACCOUNTS OR AS CASH (ORDINARY LEAST-SQUARES ESTIMATES) DEPENDENT VARIABLE: MONTHLY REMITTANCES SENT BY MIGRANT

Robust standard errors in parentheses. Significant at *10%, **5%, and ***1%. Dependent variables are monthly remittances in U.S. dollars. For the dependent variables in columns 1 and 2, all funds sent to El Salvador are counted as remittances, whether retrieved by the recipient in cash or sent directly to a bank account. Follow-up survey administered from March to June 2009. Omitted treatment indicator is for treatment 0 (comparison group). Control variables: marketer fixed effects are for the specific individual (out of nine) who conducted the marketing visit; stratification cell fixed effects for each of 48 unique combinations of stratification variables: gender (male/female), having a U.S. bank account (yes/no), relationship to remittance recipient (parent/child/spouse/other), and years in US category (0–5 years/6–10 years/11–15 years); treatment months are November 2007 through July 2008 inclusive.

impacts are obscured by the fact that the partner bank did market the Cuenta Unidos and Ahorro Directo accounts more broadly after the follow-up survey was concluded (roughly 12 months after the treatments were administered).³⁸

³⁸ While not dispositive, the time trend in savings in the comparison group (treatment 0) is suggestive that the partner bank's broad marketing of these accounts did lead savings in the comparison group to catch up with those in the other treatments. The bottom row of table 6, panel B displays mean savings in the comparison group, which show a distinct rise in the 13 to 18-month post treatment period (which appears persistent to later periods), coinciding with the timing of the partner bank's broad marketing of the new accounts.

E. Impact on Remittances

Increases in savings in El Salvador that we have documented (in response to treatment 3 in particular) could either reflect an increase in the recipient savings rate (keeping remittances constant) or, alternatively, increases in remittances sent by the migrant. We therefore examine impacts of the treatments on remittances sent by the migrant to El Salvador.

Results are presented in table 7. Panel A presents the main effects of treatments 3, 2, and 1, while panel B presents separate treatment effects for migrants with and with-

out demand for control. The dependent variable in all columns is monthly remittances sent by the migrant.

The first and second columns of the table examine migrant remittances sent via the partner bank in, respectively, the full sample and the sample of migrants completing the follow-up survey. These are remittances sent to any recipient in El Salvador (we are not able to parse out only remittances sent to the primary remittance recipient).³⁹ The results in the second column are included to facilitate comparison with the third column, which examines remittances to the primary remittance recipient via all channels (not just the partner bank), as reported by the migrant in the followup survey. For neither sample is an identifiable effect of any treatment on remittances sent via the partner bank: all coefficients in panel A are small in magnitude, and none are statistically significantly different from 0. The same conclusion holds when the dependent variable is remittances that migrants report sending to primary remittance recipients in the follow-up survey (column 3). In panel B where separate effects are estimated for migrants with and without demand for control, there also is no robust evidence of heterogeneous effects of any of the treatments on remittances.

In the context of our other findings, the lack of impact of the treatments on remittances suggests that the increases in savings we found (particularly due to treatment 3) reflect an increase in the savings rate (holding constant the flow of remittances to El Salvador).

VI. Conclusion

This paper expands our knowledge about financial decision making by international migrants, and in particular on how they respond to improvements in their ability to monitor and control financial decision making in the origin country. We implemented a field experiment that offered U.S.based Salvadoran migrants bank accounts that varied in the degree to which migrants could monitor and control savings in El Salvador-based accounts. We found that the treatment that offered migrants the greatest degree of control over El Salvador savings (offering both joint accounts and accounts in the migrant's name alone) led to substantial increases in savings at the partner bank. This increase in savings is likely due to enhanced control exerted by migrants; the effect of the treatment is significantly larger among migrants who report greater demand for such control in the baseline survey. As a caveat, we note that there are no obvious welfare implications of our results. Increased control exerted by migrants may not necessarily lead to higher well-being on the part of family members in the origin household. Migrants may be pursuing objectives that they place in higher regard than do family members back home.

Another important result of the paper is that simply channeling remittances into bank accounts in the home country does not in itself promote savings accumulation. This is clearly demonstrated by the fact that one of our treatments, which did not offer joint accounts and instead promoted opening and remitting into bank accounts in the name solely of someone in El Salvador, had no identifiable impact on savings. But when migrants are given the ability to monitor and control savings in the home country, the impact on savings accumulation is much larger. This insight should guide future efforts to facilitate savings accumulation in home country households that are connected with international migrants.

By showing the effects of an intervention that enhanced migrant control over savings in remittance-recipient households, this study also suggests some high-potential directions for future research. In particular, it should be fruitful to study the impacts of migrant control over other remittance uses that may have positive spillovers and wider development impacts, such as payments for schooling, health care, and investments in microenterprises.

REFERENCES

- Ashraf, Nava, "Spousal Control and Intra-Household Decision Making: An Experimental Study in the Philippines," *American Economic Review* 99 (2009), 1245–1277.
- Aycinena, Diego, Claudía A. Martinez, and Dean Yang, "The Impact of Remittance Fees on Remittance Flows: Evidence from a Field Experiment among Salvadoran Migrants," University of Michigan mimeograph (2010).
- Browning, Martin, and Pierre-André Chiappori, "Efficient Intra-Household Allocations: A General Characterisation and Empirical Tests," *Econometrica* 66 (1998), 1241–1278.
- Chatterjee, Promothesh, and Randall L. Rose, "Do Payment Mechanisms Change the Way Consumers Perceive Products?" *Journal of Consumer Research* 38 (2012), 1129–1139.
- Chen, Joyce, "Migration and Imperfect Monitoring: Implications for Intra-Household Allocation," *American Economic Review: Papers* and Proceedings 96 (2006), 227–231.
- Chin, Aimee, Léonie Karkoviata, and Nathaniel Wilcox, "Impact of Bank Accounts on Migrant Savings and Remittances: Evidence from a Field Experiment," University of Houston mimeograph (June 2010).
- De Laat, Joost, "Household Allocations and Endogenous Information," University of Quebec at Montreal mimeograph (September 2008).
- Dercon, Stefan, and Pramila Krishna, "In Sickness and in Health: Risk Sharing within Households in Rural Ethiopia," *Journal of Political Economy* 108 (2000), 688–727.
- Dubois, Pierre, and Ethan Ligon, "Incentives and Nutrition for Rotten Kids: Intrahousehold Food Allocation in the Philippines," University of California, Berkeley mimeograph (December 2005).
- Duflo, Esther. "Grandmothers and Granddaughters: Old Age Pension and Intra-Household Allocation in South Africa," World Bank Economic Review 17 (2003) 1–25.
- Goldstein, Markus, Alain de Janvry, and Elisabeth Sadoulet, "Is a Friend in Need a Friend Indeed? Inclusion and Exclusion in Mutual Insurance Networks in Southern Ghana," in Stefan Dercon, ed, Insurance against Poverty (New York: Oxford University Press, 2005).
- Hertzberg, Andrew, "Exponential Individuals, Hyperbolic Households," Columbia Business School working paper (2011).
- Kinnan, Cynthia, "Distinguishing Barriers to Insurance in Thai Villages," Northwestern University working paper (2011). Laran, Juliano, Amy Dalton, and Eduardo Andrade, "The Curious Case
- Laran, Juliano, Amy Dalton, and Eduardo Andrade, "The Curious Case of Behavioral Backlash: Why Brands Produce Priming Effects and Slogans Produce Reverse Priming Effects," *Journal of Consumer Research* 37 (2011), 999–1014.

³⁹ For the dependent variables in columns 1 and 2, all funds sent to El Salvador are counted as remittances, whether retrieved by the recipient in cash or sent directly to a bank account (and whether the bank account is joint with the migrant or in the name of someone in El Salvador only).

- Lundberg, Shelly, and Robert Pollak, "Separate Spheres Bargaining and the Marriage Market," *Journal of Political Economy* 101 (1993), 988–1010.
- Lusardi, Annamaria, and Olivia S. Mitchell, "Financial Literacy and Planning: Implications for Retirement Wellbeing," Pension Research Council, Wharton School, University of Pennsylvania working paper (2007).
 - "Baby Boomer Retirement Security: The Roles of Planning, Financial Literacy, and Housing Wealth," *Journal of Monetary Economics* 54 (2007), 205–224.
- Manser, Marilyn, and Murray Brown, "Marriage and Household Decision-Making: A Bargaining Analysis," *International Economic Review* 21 (1980), 31–44.
- Martinez, Claudia, "Intra-Household Allocation and Bargaining Power: Evidence from Chile," *Economic Development and Cultural Change* 61 (2013) 577–605.
- McElroy, Marjorie B., and Mary-Jean Horney, "Nash-Bargained Household Decisions: Towards a Generalization of the Theory of Demand," *International Economic Review* 22 (1981), 333–349.
- Pew Hispanic Center, *Billions in Motion: Latino Immigrants, Remittances, and Banking* (Washington, DC: Pew Hispanic Center and Multilateral Investment Fund, 2002).

- Rangel, Marcos, "Alimony Rights and Intrahousehold Allocation of Resources: Evidence from Brazil," *Economic Journal* 116 (2006), 627–658.
- Schaner, Simone, "Intrahousehold Preference Heterogeneity, Commitment, and Strategic Savings: Theory and Evidence from Kenya," Dartmouth College working paper (2011).
- Strauss, John, and Duncan Thomas, "Human Resources: Empirical Modeling of Household and Family Decisions," in Jere Behrman and T. N. Srinivasan, eds., *Handbook of Development Economics* (New York: North-Holland, 1995).
- Terry, Donald F., and Steven R. Wilson, eds., Beyond Small Change: Making Migrant Remittances Count (Washington, DC: Inter-American Development Bank, 2005).
- Udry, Christopher, "Gender, Agricultural Productivity and the Theory of the Household," *Journal of Political Economy* 104 (1996), 1010– 1046.
- World Bank, Global Economic Prospects 2006: Economic Implications of Remittances and Migration (Washington, DC: World Bank, 2006).
 ——— Close to Home: The Development Impact of Remittances in Latin
- America (Washington, DC: World Bank, 2007). Yang, Dean, "Migrant Remittances," Journal of Economic Perspectives 25 (2011), 129–152.